Author's response to reviews

Title: Pneumatized superior turbinate as a cause of headache

Authors:

Elias Homsioglou (homsioglou@hol.gr)
Dimitrios G Balatsouras (balats@panafonet.gr)
Gregory Alexopoulos (ent@yahoo.gr)
Antonis Kaberos (akaberos@hotmail.com)
Michael Katotomichealakis (michkato@freemail.gr)
Vassilios Danielides (vdaniili@med.duth.gr)

Version: 2 Date: 17 December 2006

Author's response to reviews: see over
Dear Editor,

Attached please find electronic version of our manuscript entitled “Pneumatized superior turbinate as a cause of headache” by the authors E. Homsioğlu, D. Balatsouras, G. Alexopoulos, A. Kaberos, M. Katotomichelakis and V. Danielides. The manuscript was revised according to your reviewer’s suggestions. We shall be grateful if you would re-consider it for publication in "Head & Face Medicine".

Looking forward to your answer,

Yours sincerely

Dimitris Balatsouras, M.D.
**Answers to comments of reviewer 1**

- The coronal CT figure was reversed.

**Answers to comments of reviewer 2**

- The issue of olfaction was considered (page 5, paragraph 1, lines 5-11).
- Osteomas as possible causes of headaches are now discussed (page 7, last paragraph).
- The pneumatized turbinate had not any content (fluid, debris etc.) and this is now reported (page 5, paragraph 1, lines 2-3).